Author's response to reviews

Title: Case of lupus vulgaris mimicking hemangioma diagnosed 42 years after onset

Authors:

mahizer yaldiz (drmahizer@gmail.com)
teoman m erdem (mterdem@hotmail.com)
bahar s dikicier (bsevimlidikicier@gmail.com)
fatma h dilek (fhdilek@hotmail.com)

Version: 2 Date: 21 January 2015

Author's response to reviews:

the revision that are required had been done